Pro duction of energetic electrons in the Nov em b er 27,

1997 are: a preliminary X-ray and radio analysis of

impulsive phase and sho ck wa v e acceleration

N. Vilmer

1

, K.-L. Klein

1

, J.I. Khan

2;3

, H. Aurass

4

1

DASOP

, CNRS-URA 2080, Observ atoire de Paris, Section de Meudon, F-92195

Meudon, F rance

2

Mullard Space Science Lab oratory

, University College, London, Holmbury St.

Mary

, Dorking, Surrey

, RH5 6NT, U.K

3

Mailing address: Institute of Space and Astronautical Science, 3-1-1 Y oshinodai,

Sagamihara, Kanagawa 229, Japan

4

Astrophysik alisches Institut Potsdam, An der Sternw arte 16, D-14482 Potsdam,

Germany

On Nov ember 27, 1997 a 2B/X2.6 are (1259-1338UT) was observed near the

north-eastern solar limb. It pro duced a strong hard X-ray emission from 1313 UT

to 1320 UT as well as fast drift bursts and a bright type I I burst at dm-m radio

wa velengths. In this frequency range, the radio emission can b e studied with excep-

tionally go o d sp ectral coverage and multi-frequency imaging. The p oster presents a

preliminary analysis of data from the soft and hard X-ray imagers on Y ohkoh, the

Nan� cay Radiohelio gr aph (Kerdraon & Delouis 1997) and the Potsdam-T r emsdorf

Spe ctr o gr aph (Mann et al. 1992).

W e inv estigate the spatial distribution of acceleration sites in the corona, through

a comparison of coronal plasma structures with the regions where energetic electrons

radiate radio (dm-m) emission and hard X-rays, and the temp oral evolution of this

site distribution in the course of the even t. The main results of this analysis are:

1. During the impulsive phase radio emission with di�erent frequency sp ectra

comes from di�erent lo cations within the aring active region. The radio

emission extending to the highest frequencies comes from sources near the

southern b order of the active region and is asso ciated with the hardest X-ray

emission.

2. The type I I radio sources o ccur at increasingly great distance from the ac-

tive region with decreasing frequency

, but this displacement is tangential to

the solar limb rather than radial. The sp eed inferred from the p osition mea-

surements is ab out 1400 km/s, which is twice higher than inferred from the

sp ectrographic data using a typical gravitational scale height.The �rst sig-

nature of the type I I emission o ccurs at the time and place where ejecta or

evolving lo op structures are observed by Y ohk oh-SXT.

References

Kerdraon A., Delouis J.M., 1997, in Coronal Physics from Radio and Space Ob-

serv ations, ed. G. T rottet, Springer, Lecture Notes in Physics 483, p. 192

Mann G., Aurass H., V oigt W., Pasc hke J., 1992, in Coronal Streamers, Coronal

Lo ops and Coronal and Solar Wind Comp osition, ESA SP-348, 129